Author’s response to reviews

Title: Effect of an Electronic Nicotine Delivery Device (e-Cigarette) on Smoking Cessation and Reduction: A Prospective Pilot Study

Authors:

Riccardo Polosa (polosa@unict.it)
Pasquale Caponnetto (p.caponnetto@unict.it)
Jaymin B Morjaria (J.B.Morjaria@soton.ac.uk)
Gabriella Papale (gabriellapapale@virgilio.it)
Davide Campagna (davidecampagna83@gmail.com)
Cristina Russo (kristina_russo@yahoo.com)

Version: 4 Date: 21 May 2011

Author’s response to reviews: see over
To,
Dr. Melissa Norton,
Editor-in-chief,
BMC Public Health

20th May 2011.

Dear Dr Norton,

Ref: Submission of manuscript following acceptance queries

I am writing to you with our enclosed point-by-point response to editor’s queries. We have made the some of the changes and responded to the others in an enclosed word document.

Please do not hesitate to contact us should the there be any queries. Many thanks.

Yours sincerely,

Prof Riccardo Polosa & Dr Jaymin Morjaria.